RESEARCH



A vignette study of mental health literacy for binge-eating disorder in a self-selected community sample



Kayla B. Hollett¹, Jenna M. Pennell¹ and Jacqueline C. Carter^{1*}

Abstract

Background Mental health literacy has implications for mental disorder recognition, help-seeking, and stigma reduction. Research on binge-eating disorder mental health literacy (BED MHL) is limited. To address this gap, our study examined BED MHL in a community sample.

Method Two hundred and thirty-five participants completed an online survey. Participants read a vignette depicting a female character with BED then completed a questionnaire to assess five components of BED MHL (problem recognition, perceived causes, beliefs about treatment, expected helpfulness of interventions, and expected prognosis).

Results About half of participants correctly identified BED as the character's main problem (58.7%). The most frequently selected cause of the problem was psychological factors (46.8%) and a majority indicated that the character should seek professional help (91.9%). When provided a list of possible interventions, participants endorsed psychologist the most (77.9%).

Conclusions Compared to previous studies, our findings suggest that current BED MHL among members of the public is better, but further improvements are needed. Initiatives to increase knowledge and awareness about the symptoms, causes, and treatments for BED may improve symptom recognition, help-seeking, and reduce stigma.

Plain English summary

Ensuring that people have accurate knowledge about eating disorders is important to reduce stigma and improve access to treatment. We conducted a study to explore what people know about binge-eating disorder mental health literacy (BED MHL). Two hundred and thirty-five participants read a vignette about an adult woman with BED and then completed an online survey to assess their knowledge. Most participants correctly identified BED as the main problem and identified appropriate causes and treatments. Our results help clarify what people know and what they do not know about BED, which can inform programs that are designed to reduce stigma and improve access to care.

Introduction

*Correspondence: Jacqueline C. Carter

jacquelinec@mun.ca

¹ Department of Psychology, Memorial University of Newfoundland, 230 Elizabeth Avenue, St. John's, NL A1C 5S7, Canada



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Mental health literacy (MHL) refers to public knowledge and understanding about mental disorders [1]. Studies show that increased MHL is associated with improved

problem recognition, better help-seeking attitudes, and

reduced stigma [2-4]. As such, increasing MHL is an

important public health priority. Moreover, MHL is

important for stigma reduction because it represents

a policy variable that can be targeted directly to alter stigma [5]. To date, considerable research has focused on the consequences of mental disorder stigma, but less research has examined MHL for mental disorders, and research on MHL for eating disorders is particularly lacking [6]. Among the most common eating disorders, binge eating disorder (BED) has received the least MHL research attention [7].

BED is a serious eating disorder characterized by recurrent episodes of binge eating in the absence of extreme weight control behaviour such as purging [8]. Episodes of binge eating occur when an individual consumes an objectively large amount of food for the context, typically in a discrete time frame, while experiencing a sense of loss of control [8]. The disorder is associated with significant distress, high rates of psychiatric comorbidity [9–11], medical problems [12], and poor quality of life [13, 14]. While BED is the most common eating disorder affecting at least 2.8% of the population [11, 15], it remains underrecognized and undertreated [16] with up to 90% of people not receiving evidence-based treatment [11, 17]. Moreover, individuals with BED face considerable stigma in that they tend to be perceived as lacking self-discipline and are often blamed for their condition [18-20].

Research to date has provided preliminary evidence that the public have poor BED MHL. For example, O'Connor and colleagues [21] found that BED was the only condition of several mental and physical conditions depicted in vignettes that all participants failed to identify using an open-ended question, indicating a low ability to recognize the disorder (N=290 adolescents). In another study, Anderson and colleagues [22] found that 61.9% of a sample of adolescents (N=1666) were unable to correctly identify BED in a vignette and also viewed BED as less severe than bulimia nervosa. Further, McNicholas and colleagues [23] found that 80.8% of healthcare professionals failed to recognize BED in a vignette and were less likely to identify BED than bulimia nervosa, major depressive disorder, and type 1 diabetes. Taken together, these findings suggest that adolescents and healthcare professionals are not good at identifying BED from written descriptions.

In a study conducted with the general population, Mond and Hay [24] found that 88.3% of participants failed to identify BED as the correct diagnosis for a vignette character. Moreover, participants endorsed "Getting out and about more/finding some new hobbies" as the most suitable treatment option for the character's problem. This study had some important strengths, namely the use of a nationally representative sample of 1034 participants aged 15–94. However, a limitation of the study was that the vignette character was described as "severely obese", which may have influenced participant responses given that obesity is highly stigmatized [25, 26]. Second, the vignette description included details not pertinent to a diagnosis of BED (e.g., thoughts of engaging in weight control through exercise or laxatives). Vignette descriptors that are not diagnostic features of BED may have distracted participants from important information such as the description of the character's binge eating episodes. Finally, considerable time has passed since Mond and Hay [24] published their work in 2008, and BED has since received recognition in the Diagnostic and Statistical Manual of Mental Disorders, 5th edition [8]. Therefore, it is possible that public knowledge about BED has changed, and a current assessment of BED MHL among members of the public is needed.

The aim of our study was to assess current BED MHL using a vignette-survey design while addressing some limitations of previous studies. Namely, we based our vignette description of BED on DSM-5 diagnostic criteria while excluding extraneous information that could distract from BED recognition (e.g., describing the character's body weight or unrelated dieting behaviour). Additionally, we updated previous MHL questionnaires to better assess domains of MHL relevant for BED including participants' ability to identify the main problem, their knowledge about likely causes of the problem, their understanding of the need for treatment, their ability to identify helpful interventions, and their ability to anticipate a likely prognosis if treatment were received.

Additionally, studies suggest that sociodemographic factors such as gender, age, level of education, and personal experience with an eating disorder may influence problem recognition and opinions about help-seeking for eating disorders [20, 22, 24]. Based on preliminary findings from previous studies on eating disorders, we hypothesized that being female and being younger in age would be associated with improved ability to correctly identify BED in the vignette [22, 24, 27]. We also hypothesized that being female, having a higher level of education, and having personal experience with an eating disorder would be associated with participants' recommendation that the vignette character should seek professional help [20, 22, 24, 28, 29].

Method

Participants and procedure

Two-hundred and forty-four participants were recruited through advertisements posted on social networking sites (e.g., Facebook groups in Canada) and in public locations throughout St. John's, Canada. The advertisements indicated that participants would complete an online survey on "opinions about eating behaviours" and did not mention BED or MHL specifically. Participants accessed the survey via Qualtrics by clicking on a link or emailing a research assistant. Upon accessing the survey, participants were presented with a consent form, demographic questions, a vignette, a MHL questionnaire, and a feedback sheet. Participants were excluded if they were less than 18 years old (n=3), spent less than seven seconds on the vignette screen (n=4), or did not complete any MHL questions (n=2). The final sample (N=235)was 85.1% female with a mean age was 37.6 years (range: 18-75 years). Most participants identified as Caucasian (95.7%), followed by Indigenous (2.6%), and other (1.6%). Most completed some post-secondary education (87.5%), and 8.1% self-reported having a prior eating disorder diagnosis. Participants were invited to enter a draw for one of two \$50 gift cards as a token of appreciation. This study was approved by the Memorial University Interdisciplinary Committee on Ethics in Human Research.

Materials

A vignette was developed with reference to best practice recommendations [30, 31] and previous vignettes used in eating disorder research [18, 20, 24, 32]. The vignette (see Additional file 1) described a 30-year-old woman named Jane who had BED as per DSM-5 criteria [8]. The vignette did not mention Jane's weight, the term "binge eating", or reveal that she had a diagnosis of BED.

The MHL questionnaire (see Additional file 1) was modelled from previous MHL questionnaires [24, 33-36]. The first question was, "Please indicate which weight category you think Jane likely belongs to" to assess beliefs about weight associated with binge eating. MHL questions followed to assess: (1) ability to identify the main problem; (2) perceived causes of the problem; (3) perceived need for treatment; (4) perceived helpfulness of interventions; and (5) perceived prognosis. The questions were ordered from general (e.g., "Do you think Jane has a problem?") to specific (e.g., "What do you think is the primary cause of Jane's problem?"). Following this, participants were asked if they currently or have previously had a problem similar to the character. Lastly, participants were informed that the character had BED and were then asked several direct questions (e.g., "Do you think bingeeating disorder is an official mental disorder?"). For each MHL question, participants were given several options to select from including, "Other (please specify)", "Not sure", "None of the above", and "Not applicable (I do not think Jane has a problem)" to limit biased responding.

Statistical analyses

Analyses were conducted using Jamovi Version 1.1.9.0. Descriptive statistics were used to assess responses to the MHL questions. Chi Squared tests of independence and Pearson correlations were used to examine relationships
 Table 1
 Participant responses to, "How would you describe

 Jane's problem?"

 Mental health problem	51.1%
Both a physical and mental health problem	42.6%
Other	3.8%
Physical health problem	1.7%
Don't think Jane has a problem	0.9%

 Table 2
 Participant responses to, "What would you say is Jane's main problem?"

Binge eating disorder	58.7%
Not sure	13.6%
An eating disorder, but not anorexia or bulimia	6.8%
Depression	6.4%
Anxiety	4.7%
Other	4.3%
Bulimia nervosa	3.0%
Obesity/overweight	1.7%
Anorexia nervosa	0.9%

between demographic variables and participants' ability to identify BED as well as their help-seeking recommendations for the vignette character. A Fischer's exact test was conducted (due to an expected cell frequency < 5) to examine the relationship between personal experience with an eating disorder and opinions about help-seeking. A Bonferroni-corrected alpha level of 0.01 (0.05/5) was used for these analyses.

Results

Perceived character weight

When asked, "Please indicate which weight category you think Jane likely belongs to", the vast majority of participants characterized Jane as either overweight or obese (90.1%) followed by the recommended weight range (7.8%), and the underweight weight range (2.2%).

Mental health literacy

When asked "Do you think Jane has a problem?" most participants responded "Yes" (88.9%), followed by "Not sure" (10.2%), and "No" (0.9%). When asked, "How would you describe Jane's problem?", about half of the sample (49.5%) responded that Jane had a mental health problem (see Table 1). Next, when asked "What would you say is Jane's *main* problem?", more than half of participants (58.7%) selected BED (see Table 2). To assess perceived causes participants were asked, "What do you think is the *primary cause* of Jane's problem?" and the most common response was "Psychological factors" (46.8%), followed by "Not sure" (26.8%) (see Table 3). Regarding perceived need for treatment, participants were asked, "Do you think that Jane should seek professional help for her problem?" and the majority of participants selected "Yes" (91.9%). When asked to indicate which person-based, therapy-based, and medicationbased interventions they believed would be most helpful, the most common response for each category was, "Psychologist" (77.9%), "Other psychotherapy" (50.2%), and "Not sure" (35.3%), respectively (see Table 4 for the top five responses in each category). Regarding the perceived helpfulness of receiving treatment, participants were asked, "What do you think Jane's likely prognosis would be if she received treatment?" and the most common response was "She would have a full recovery, but problems might reoccur" (70.6%) (see Table 5).

Participants' experience of a similar problem

When asked, "In the past, have you experienced something similar to Jane's problem?", over half the sample indicated that they believed they have experienced a similar problem to Jane in the past (54.9%). When asked, "Do you believe you might currently have a problem similar to Jane's problem?", nearly a third of participants reported that they believe they currently have a problem similar to Jane's (32.9%).

After being informed that Jane has BED, participants were asked, "Have you ever heard of binge-eating disorder?" and the vast majority of participants responded "Yes" (95.3%). When asked "Do you think binge-eating disorder is an official mental disorder?", participants selected "Yes" most often (77.4%), followed by "Not sure" (20.4%), and "No" (2.1%). When asked, "How common do you think binge-eating disorder is in the general adult population?" 42.2% of participants indicated more than 15% of the population, 40% of participants indicated between 5 and 15% of the population, and 17.9% of participants indicated 5% or less of the population.

Table 3 Participant responses to, "What do you think is the primary cause of Jane's problem?"

Psychological factors (e.g., self-esteem)	46.8%
Not sure	26.8%
Social factors (e.g., upbringing, life stressors)	20.4%
Other (please specify) ^a	5.1%
Biological factors (e.g., genetics)	0.9%

^a Participants who selected "Other" were asked to specify. There were twelve responses, nine of which suggested that a combination of psychological, social, and biological factors contributed to Jane's problem, representing 3.8% of the total sample.

Table 4 Top five participant responses to, "If Jane did receive help, which of the following [person-based, therapy-based, medication-based] interventions do you believe would be most helpful?"

Person-based	
Psychologist	77.9%
Self-help support group	48.9%
Dietician	46.4%
Family doctor	32.8%
Family member or close friend	18.7%
Therapy-based	
Other psychotherapy	50.2%
Behavioural weight loss and/or exercise program	44.7%
Cognitive behavioural therapy	42.6%
Family counselling/therapy	38.7%
Using a self-help treatment manual or self-help book	25.1%
Medication-based	
Not sure	35.3%
Antidepressant medication	29.4%
Vitamins and minerals	28.1%
Antianxiety medication	24.3%
Herbal medicines/tonics	13.6%

Sociodemographic variables and mental health literacy

Females were approximately twice as likely than males to correctly identify BED in the vignette (61% correct among females vs. 45% correct among males), but this difference was not significant, χ^2 (1, n=231)=2.781, p = .095, OR = 1.899. Additionally, females were approximately twice as likely as males to recommend that the character seek professional help (93% recommendation among females vs. 87% among males), but this difference was also not significant, χ^2 (229, 1)=1.259, p=.2618, OR = 1.947. Regarding age, there was a significant moderate negative correlation between age and participants' ability to correctly identify BED in the vignette, r = -.257, p < .01, indicating that younger participants were better able to recognize BED in the vignette than older participants. Regarding education, there was no significant correlation between level of education and participants' belief that the character should seek professional help, r = .063, p = .340. Regarding personal experience,

 Table 5
 Participant responses to, "What do you think Jane's likely prognosis would be if she received an appropriate treatment?"

- Full recovery, but problems might reoccur	70.6%
Not sure	18.3%
Partial recovery	5.5%
Full recovery with no further complications	5.1%
Don't think Jane has a problem	0.4%

participants with personal experience were approximately nine times more likely to recommend that the character seek professional help compared to participants without personal experience (100% recommendation among those with personal experience vs. 90.1% among those without), although this difference was not statistically significant, p = .051, OR = 9.330.

Discussion

This study assessed BED MHL among a self-selected sample of adults from the general population, primarily females. Overall, we found better BED MHL in our sample compared to previous studies. In terms of problem recognition, 58.7% of participants successfully identified BED as the most likely problem, which contrasts with lower problem recognition rates of 11.7% and 19.2% observed in prior studies [23, 24]. These differences could be explained by methodological differences across studies, such as differences in vignettes, or they could be explained by differences in study samples. For example, we did not mention the character's weight in our vignette, making it less likely for participants to confuse BED with obesity. Also, about one third of our participants reported having a problem similar to the vignette character, which is a high proportion compared to previous samples [24]. It could be the case that individuals with personal experience with binge eating were drawn to participate in our study, which was advertised as a study on opinions about eating behaviours. In turn, the sample we recruited may have been more likely to identify and have knowledge about BED. Moreover, we found that being younger and female were associated with better BED recognition. While previous findings on age and BED recognition are conflicting [24, 27, 28], studies have found that females tend to be more knowledgeable about mental disorders [22, 37]. Taken together, these findings suggest that BED MHL initiatives could target groups of people who many have less knowledge on BED (e.g., people with less education, older adults, males). However, more research is needed to better understand how members of the general public understand BED, and researchers should continue to explore relationships between BED MHL and demographic characteristics to better understand who would benefit most from MHL initiatives.

Regarding perceived causes, 46.8% of participants selected "Psychological factors" as the primary cause of the character's problem, suggesting that many understood that BED is related to mental well-being [38, 39]. However, 26.8% of participants selected "Not sure", suggesting that many were not able to characterize BED as a mental health problem. Interestingly, participants selected "Biological factors" as the least likely cause (0.9%), which is consistent with previous findings [40].

However, the belief that biological factors do not play a causal role in the development of BED could reflect the misconceptions that people with BED are to blame for their behaviour and lack self-discipline [18, 20]. These findings point to the importance of highlighting the biopsychosocial nature of BED in MHL initiatives.

Regarding knowledge about help-seeking and interventions, most participants (91.9%) agreed that the vignette character should seek professional help. Additionally, most (70.6%) reported that if the vignette character did receive treatment, "She would have a full recovery, but problems might reoccur". These findings are indicative of good MHL given that, while BED symptoms improve with treatment [41], relapse can occur [42]. In terms of interventions, most participants reported that seeing a psychologist (77.9%) would be most helpful person-based intervention for the character. In line with this, participants commonly selected psychotherapies such as cognitive behavioural therapy (42.6%) and family counselling (38.7%) as suitable interventions [41]. While we do not expect most members of the public to be highly familiar with BED treatments, BED MHL initiatives can include information on what is an appropriate versus an inappropriate intervention.

In terms of demographic variables related to help-seeking knowledge, females were two times more likely than males to recommend that the character seek professional help and participants with personal experience were nine times more likely than those without personal experience. Despite these moderate to large effect sizes, these differences were not statistically significant, which could reflect a lack of statistical power given our sample of 235 participants. Research would benefit from an in-depth examination of the sociodemographic characteristics related to BED MHL in a larger sample.

Given that our sample was largely female with a majority endorsing a similar problem at some point in their life, it is both surprising and encouraging that most participants agreed that the vignette character should seek help for her problem. In a vignette study by Mond and colleagues [43] on MHL for bulimia nervosa, female participants who were high risk or symptomatic for bulimia were more likely to indicate that they would not seek help if they had a problem like the vignette character because they would not want anyone to know, and symptomatic participants were more likely to believe that someone with bulimia would face discrimination. In contrast, our finding on help-seeking suggests that most people view binge eating as a valid health problem warranting treatment. In fact, individuals with lived experience in our sample were nine times more likely to recommend help-seeking. Today, BED MHL initiatives should incorporate the voices of individuals with lived experience to

encourage help-seeking and reduce stigma associated with this disorder.

Strengths, limitations, and future directions

To our knowledge, this is the first study to examine BED MHL since 2008 [24]. A primary strength was that our vignette did not include mention of the character's body size, which reduced the possibility that weight stigma impacted our results [44, 45]. We also used DSM-5 criteria in our description and excluded symptoms that do not characterize BED (e.g., thoughts of compensatory behaviours). Additionally, we examined previous MHL questionnaires and removed options less relevant to eating behaviour (e.g., "Assertiveness or social skills training" was removed as a treatment option).

While our study provides a timely update to the existing BED MHL literature, there are some limitations. First, our sample consisted primarily of white, female participants. Future studies should recruit samples that are more diverse in gender and ethnicity/race, or participants that are underrepresented in this literature (e.g., males). Related to this, more than half of our sample (54.9%) reported having experienced a similar problem in the past, and one third (32.9%) reported currently having a similar problem. These percentages could reflect the fact that many people experience difficulties with overeating [10] or a tendency for individuals with eating difficulties to participate in studies about eating. This imposes an important restriction on the generalizability of our findings as our sample may have more knowledge about eating disorders than a random sample. Additionally, our use of a female vignette character could be considered a limitation because a significant proportion of individuals with BED are male [10, 16]. While we opted to use a female character to allow for comparisons with existing studies, responses to MHL questions could vary with character gender. More research is needed to examine vignette character demographic characteristics in relation to BED MHL. Lastly, the use of a multiple-choice format in our MHL questionnaire could have increased the likelihood of biased responding from participants. Future studies would benefit from using more openended questioning in questionnaires designed to assess MHL.

Conclusions

Our study provides updated findings on public knowledge about BED. Understanding what people do not know about BED is essential for the development of effective MHL initiatives. BED MHL initiatives are necessary to improve problem recognition, help-seeking, and access to effective treatments [2, 4] as well as to reduce the stigmatization of BED [2, 3].

Abbreviation

BED MHL Binge-eating disorder mental health literacy

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40337-023-00795-y.

Additional file 1. MHL vignette and questionnaire.

Acknowledgements

None.

Author contributions

All listed authors meet the ICMJE criteria. All authors contributed significantly to the creation of this manuscript. Study conception and design: KH, JC, and JP. Acquisition of data and data analysis: KH and JP. Analysis and interpretation of data: KH, JP, and JC. Drafting of manuscript: JP, KH, and JC. Editing and corrections: KH, JC, and JP. All authors read and approved the final manuscript.

Funding

This was a non-funded study.

Availability of data and materials

The data and materials supporting the conclusions of this article are available from the authors upon reasonable request.

Declarations

Ethics approval and consent to participate

This study was approved by the Interdisciplinary Research Ethics Board at Memorial University of Newfoundland and research participants provided informed consent.

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Received: 12 April 2023 Accepted: 21 April 2023 Published online: 04 May 2023

References

- Jorm AF. Mental health literacy: public knowledge and beliefs about mental disorders. Brit J Psychiat. 2000;177:396–401.
- Cheng H, Wang C, Mcdermott RC, Kridel M, Rislin JL. Self-stigma, mental health literacy, and attitudes toward seeking psychological help. J Couns Dev. 2018;96:64–74.
- Milin R, Kutcher S, Lewis SP, et al. Impact of a mental health curriculum on knowledge and stigma among high school students: a randomized controlled trial. J Am Acad Child Adolesc Psychiatry. 2016;55:383–91.
- Rusch N, Evans-Lacko SE, Henderson C, Flach C, Thornicroft G. Knowledge and attitudes as predictors of intentions to seek help for and disclose a mental illness. Psychiatr Serv. 2011;62:675–8.
- Pescosolido BA, Martin JK, Lang A, Olafsdottir S. Rethinking theoretical approaches to stigma: a framework integrating normative influences on stigma (FINIS). Soc Sci Med. 2008;67:431–40.
- Mond J. Eating disorders "mental health literacy": an introduction. J Ment Health. 2014;23:51–4.
- Bullivant B, Rhydderch S, Griffiths S, Mitchison D, Mond JM. Eating disorders "mental health literacy": a scoping review. J Ment Health. 2020;29:336–49.

- American Psychiatric Association. Diagnostic and statistical manual of mental disorders, 5th ed. Arington; 2013.
- Brownley KA, Berkman ND, Peat CM, et al. Binge-eating disorder in adults: a systematic review and meta-analysis. Ann Intern Med. 2016;165:409–20.
- Hudson JI, Hiripi E, Harrison GP, Kessler RC. The prevalence and correlates of eating disorders in the national comorbidity survey replication. Biol Psychiatry. 2013;61:348–58.
- Kessler RC, Berglund PA, Chiu WT, et al. The prevalence and correlates of binge-eating disorder in the World Health Organization world mental health surveys. Biol Psychiatry. 2013;73:904–14.
- Mitchell JE, King WC, Pories W, Wolfe B, Flum DR, Spaniolas K. Binge eating disorder and medical comorbidities in bariatric surgery candidates. Int J Eat Disord. 2015;48:471–6.
- Pawaskar M, Witt EA, Supina D, Herman BK, Wadden TA. Impact of binge-eating disorder on functional impairment and work productivity in an adult community sample in the United States. Int J Clin Pract. 2017;71:1–9.
- 14. Reas DL. Public and healthcare professionals' knowledge and attitudes toward binge-eating disorder: a narrative review. Nutrients. 2017;9:1–17.
- Udo T, Grilo CM. Prevalence and correlates of DSM-5-defined eating disorder in a nationally representative sample of U.S. adults. Biol Psychiatry. 2018;84:345–54.
- 16. Guerdjikova Al, Mori N, Casuto LS, Mcelroy SL. Binge-eating disorder. Psychiatr Clin North Am. 2017;40:255–66.
- Sonneville KR, Lipson SK. Disparities in eating disorder diagnosis and treatment according to weight status, race/ethnicity, socioeconomic background, and sex among college students. Int J Eat Disord. 2018;51:518–26.
- Ebneter D, Latner JD. Stigmatising attitudes differ across mental health disorders: a comparison of stigma across eating disorders, obesity, and major depressive disorder. J Nerv Ment. 2013;201:281–5.
- Evans CM. Suffering in silence" the experience of treatment seeking for binge-eating disorder: an interpretative phenomenological analysis. Staffordshire University. Proquest Dissertations Publishing. 2016. https://bit. ly/2UEJCJd.
- 20. Simpson CC, Mazzeo SE. Attitudes toward orthorexia nervosa relative to DSM-5 eating disorders. Int J Eat Disord. 2017;50:781–92.
- O'Connor C, McNamara N, O'Hara L, McNicholas F. Eating disorder literacy and stigmatising attitudes towards anorexia, bulimia and binge-eating disorder among adolescents. Adv Eat Disord. 2016;4:125–40.
- Anderson R, Gratwick-Sarll K, Bentley C, Harrison C, Mond J. Adolescents' perception of the severity of binge-eating disorder: a population-based study. J Ment Health. 2015;25:16–22.
- McNicholas F, O'Connor C, O'Hara L, McNamara N. Stigma and treatment of eating disorders in Ireland: healthcare professionals' knowledge and attitudes. Ir J Psychol Med. 2016;33:21–31.
- Mond JM, Hay PJ. Public perceptions of binge eating and its treatment. Int J Eat Disord. 2008;41:419–26.
- 25. Puhl RM, Brownell KD. Psychosocial origins of obesity stigma: toward changing a powerful and pervasive bias. Obes Rev. 2003;4:213–27.
- Puhl RM, Heuer CA. The stigma of obesity: a review and update. Obesity. 2009;17:941–64.
- Smith MC, Pruitt JA, Mann LM, Thelen MH. Attitudes and knowledge regarding bulimia and anorexia nervosa. Int J Eat Disord. 1986;5:545–53.
- Mond J, Hay P, Rodgers B, Owen C, Beumont P. Beliefs of the public concerning the helpfulness of interventions for bulimia nervosa. Int J Eat Disord. 2004;36:62–8.
- Rodgers RF, Paxton SJ, McLean SA. Stigmatizing attitudes and beliefs toward bulimia nervosa: the importance of knowledge and eating disorder symptoms. J Nerv Ment. 2015;203:259–63.
- Aguinis H, Bradley KJ. Best practice recommendations for designing and implementing experimental vignette methodology studies. Organ Res Methods. 2014;17:351–71.
- 31. Huges R, Huby B. The construction and interpretation of vignettes in social science research. Soc Work Soc Sci Rev. 2004;11:36–51.
- Murakami JM, Essayli JH, Latner JD. The relative stigmatization of eating disorders and obesity in males and females. Appetite. 2016;102:77–82.
- Mond JM, Marks P, Hay PJ, et al. Mental health literacy and eating-disordered behaviour: beliefs of adolescent girls concerning the treatment of and treatment-seeking for bulimia nervosa. J Youth Adolesc. 2007;36:753–62.

- 34. Mond JM, Hay PJ, Rodgers B, Owen C. Mental health literacy and eating disorders: what do women with bulimic eating disorders think and know about bulimia nervosa and its treatment? J Ment Health. 2008;17:565–75.
- Garcia-Soriano G, Roncero M. What do spanish adolescents think about obsessive-compulsive disorder? Mental health literacy and stigma associated with symmetry/order and aggression-related symptoms. Psychiatry Res. 2017;250:193–9.
- Kim JE, Saw A, Zane N. The influence of psychological symptoms on mental health literacy in college students. Am J Orthopsychiatr. 2015;85:620–30.
- Cotton SM, Wright A, Harris MG, Jorm AF, Mcgorry PD. Influence of gender on mental health literacy in young Australians. Aust N Z J Psychiatry. 2016;40:790–6.
- Harrison C, Mitchison D, Rieger E, Rodgers B, Mond J. Emotion regulation difficulties in binge-eating disorder with and without the overvaluation of weight and shape. Psychiatry Res. 2016;245:436–42.
- Leehr EJ, Krohmer K, Schag K, Dresler T, Zipfel S, Giel KE. Emotion regulation model in binge-eating disorder and obesity: a systematic review. Neurosci Biobehav Rev. 2015;49:125–34.
- Ebneter DS, Latner JD, O'Brien KS. Just world beliefs, causal beliefs, and acquaintance: associations with stigma toward eating disorders and obesity. Pers Individ Dif. 2011;51:618–22.
- Hilbert A, Petroff D, Herpertz S, et al. Meta-analysis of the efficacy of psychological and medical treatments for binge-eating disorder. J Consult Clin Psychol. 2019;87:91–105.
- 42. Linardon J. Rates of abstinence following psychological or behavioral treatments for binge-eating disorder: meta-analysis. Int J Eat Disord. 2018;51:785–97.
- Mond JM, Hay PJ, Paxton SJ, et al. Eating disorders "mental health literacy" in low risk, high risk and symptomatic women: implications for health promotion programs. Eat Disord. 2010;18:267–85.
- Hollett KB, Carter JC. Separating binge-eating disorder stigma and weight stigma: a vignette study. Int J Eat Disord. 2021;54:755–63.
- 45. Star A, Hay P, Quirk F, Mond J. Perceived discrimination and favourable regard toward underweight, normal weight and obese eating disorder sufferers: implications for obesity and eating disorder population health campaigns. BMC Obes. 2015;2:1–9.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

